Author's response to reviews

Title: Multiple cavernous hemangiomas of the skull with dural tail sign: A case report and literature review

Authors:

Peng Xu (xupenga911@163.com)
Shengyong Lan (gxlansyu@163.com)
Youming Liang (liang_youming@sohu.com)
Quan Xiao (xiaobsh@yahoo.com.cn)

Version: 3 Date: 15 August 2013

Author's response to reviews: see over
Dear Josef:

On behalf of my co-authors, we thank you very much for giving us an opportunity to revise our manuscript, we appreciate editor and reviewers very much for their positive and constructive comments and suggestions on our manuscript entitled “Multiple cavernous hemangiomas of the skull with dural tail sign: A case report and literature review”. (MS: 9170896549569190).

We have studied reviewer’s comments carefully and have made revision which marked in red in the paper. We have tried our best to revise our manuscript according to the comments. Attached please find the revised version, which we would like to submit for your kind consideration.

We would like to express our great appreciation to you and reviewers for comments on our paper. Looking forward to hearing from you.

If you have any queries, please don’t hesitate to contact me.

Thank you and best regards.

Yours sincerely,

Peng Xu, M.D.
Department of Neurosurgery, The People's Hospital of Guangxi Zhuang Autonomous Region, Nanning,, 530021, P.R.China.
Phone/Fax:  +86-771-2186531, Email: Xupenga911@163.com